Author's response to reviews

Title: Insertion of a foldable hydrophobic IOL through the trabeculectomy fistula in cases with Microincision cataract surgery combined with trabeculectomy.

Authors:

Tanuj Dada Dr (tanujdada@hotmail.com)
Rajamani Muralidhar Dr (muralidharr@yahoo.com)
Harinder S Sethi Dr (sethi2211@yahoo.com)

Version: 4 Date: 18 December 2005

Author's response to reviews: see over
To, □ □ □ □ □ □ □ □ □ 18□

□Dec,□2005

The Editor,

BMC ophthalmology

Subject : submission of the final revised manuscript formatting changes for

MS: [8789154587212700]

Dear Sir,

Please find enclosed manuscript titled “Insertion of a foldable hydrophobic IOL through the trabeculectomy fistula in cases with Microincision cataract surgery combined with trabeculectomy” for publication under “Technical advance” section of your esteemed journal. The various formatting changes made to the revised manuscript are as follows.

1. Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.

Ans. these are identical in the manuscript and in the submission system.

2. Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Ans. Done

3. Underlining - Please do not underline any section headings.

Ans The underlining has been removed
4. **Qualifications** - Please remove the authors' qualifications (e.g. PhD, M.D.) from the title page.

   **Ans.** Done

5. **Competing interests** - Please include a ‘Competing interests’ section between the List of Abbreviations [if any, change to Conclusions if not] and Authors’ contributions. If there are none to declare, please write ‘The authors declare that they have no competing interests’.

   Ans. The above mentioned changes have been incorporated

6. **Keywords section** - Please remove the keywords section from the manuscript.

   Ans. The keywords section from the manuscript has been removed

7. **Author's contributions** - Please change to Authors' contributions.

   Ans. Done


Thanking you,


Sincerely yours,


**Harinder Singh Sethi, MD, FRCS**